Author's response to reviews

Title: Compensatory quadrant-hyperhidrosis after contralateral intrathoracic surgery: a case report

Authors:

Stefan Brodoehl (stefan.brodoehl@med.uni-jena.de)
Otto OW Witte (otto.witte@med.uni-jena.de)
Albrecht Guenther (albrecht.guenther@med.uni-jena.de)

Version: 7 Date: 6 November 2012

Author's response to reviews:

Revision of a Case report in Journal of Medical Case Reports
MS: 1721377209793337

Dear Editor,

Thank you for providing us with the opportunity to revise this manuscript. As written in your last E-mail (06.11.12) we moved the patient’s details from the Introduction to the Case presentation section. Instead we moved the first paragraph of the Discussion to the Introduction. We also added a short Abbreviation section. As suggested by one of our reviewers we added a table summarizing recent literature concerning unilateral hyperhidrosis; therefore we apologize for extending the number of references. To increase the anonymity of our patient we enlarged the blackening box.

We sincerely hope that you will find our revised manuscript to be acceptable for publication in Journal of Medical Case Reports.

Yours sincerely,

Stefan Brodoehl, M.D. Otto W. Witte, M.D. Albrecht Guenther, M.D.